Author's response to reviews

Title: Internal sphincterotomy reduces postoperative pain after Milligan Morgan haemorrhoidectomy.

Authors:

Giuseppe Diana (gadiana@unipa.it)
Giovanni Guercio (giangu@unipa.it)
Bianca Cudia (biancacudia@unipa.it)
Calogero Ricotta (calogeroricotta@yahoo.com)

Version: 5 Date: 12 October 2009

Author's response to reviews:

Palermo 11.10.2009

Dear Sir,

We revised, as requested, our manuscript by performing:

1. Revision of the entire manuscript
2. Correction of the order of the references
3. Providing place of publication in “reference 33”
4. Revision of the tables and formatting “Table 3”
5. Indication of the country in the affiliation details

All authors have read the revised manuscript and agreed to its content.

With best regards
(Prof. Giuseppe Diana)